

Citation: Lau M-J, Ross PA, Endersby-Harshman NM, Yang Q, Hoffmann AA (2022) *Wolbachia* inhibits ovarian formation and increases blood feeding rate in female *Aedes aegypti*. PLoS Negl Trop Dis 16(11): e0010913. https://doi.org/ 10.1371/journal.pntd.0010913

Editor: Richard A. Bowen, Fort Collins, UNITED STATES

Received: August 1, 2022

Accepted: October 28, 2022

Published: November 11, 2022

Copyright: © 2022 Lau et al. This is an open access article distributed under the terms of the <u>Creative</u> Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All data underlying the findings in this paper are fully available without restriction at https://melbourne.figshare.com/ articles/dataset/Fertility_paper_Lau_et_al_2022/ 21357654.

Funding: AAH was funded by the National Health and Medical Research Council (1132412, 1118640, www.nhmrc.gov.au). The funders had no role in the study design, data collection and analysis, decision to publish, or preparation of the manuscript. RESEARCH ARTICLE

Wolbachia inhibits ovarian formation and increases blood feeding rate in female *Aedes aegypti*

Meng-Jia Lau*, Perran A. Ross, Nancy M. Endersby-Harshman, Qiong Yang, Ary A. Hoffmann [®]*

Pest and Environmental Adaptation Research Group, Bio21 Institute and the School of BioSciences, The University of Melbourne, Parkville, Victoria, Australia

* lau.mengjia@gmail.com (MJL); ary@unimelb.edu.au (AAH)

Abstract

Wolbachia, a gram-negative endosymbiotic bacterium widespread in arthropods, is wellknown for changing the reproduction of its host in ways that increase its rate of spread, but there are also costs to hosts that can reduce this. Here we investigated a novel reproductive alteration of *Wolbachia wAlbB* on its host *Aedes aegypti*, using studies on mosquito life history traits, ovarian dissection, as well as gene expression assays. We found that an extended period of the larval stage as well as the egg stage (as previously shown) can increase the proportion of *Wolbachia*-infected females that become infertile; an effect which was not observed in uninfected females. Infertile females had incomplete ovarian formation and also showed a higher frequency of blood feeding following a prior blood meal, indicating that they do not enter a complete gonotrophic cycle. Treatments leading to infertility also decreased the expression of genes related to reproduction, especially the vitellogenin receptor gene whose product regulates the uptake of vitellogenin (Vg) into ovaries. Our results demonstrate effects associated with the development of infertility in *w*AlbB-infected *Ae. aegypti* females with implications for *Wolbachia* releases. The results also have implications for the evolution of *Wolbachia* infections in novel hosts.

Author summary

Wolbachia bacteria reduce the transmission of dengue and other arboviruses transmitted by their mosquito hosts once they have invaded host populations. This invasion process is being undertaken in *Aedes aegypti* mosquito populations throughout the world but can be slowed by deleterious fitness costs that the *Wolbachia* bacteria have on their mosquito hosts. Here we investigate in detail a recently identified cost where one of the *Wolbachia* strains being used in invasions causes infertility in females when the females develop from eggs that have been stored for a period. We show that slowed larval development (through food reduction) can cause the same phenomenon and that the emerged females lack functional ovaries. These infertile females also take blood meals more often. The results have implications for the successful invasion of *Wolbachia* under ecological circumstances **Competing interests:** The authors have declared that no competing interests exist.

where the mosquito eggs have to persist in the absence of water for a period and where larval food is limiting.

Introduction

Wolbachia, a *Rickettsia*-like maternally inherited endosymbiotic bacterium widespread in insects [1], can alter insect reproduction in multiple ways to enhance its transmission [2]. Specific alterations include parthenogenesis–infected females can produce female offspring without mating; feminization–infected genetic males are transformed into functional females and produce female offspring; male killing–infected immature males die while females survive, and cytoplasmic incompatibility (CI)–females infected with the predominant *Wolbachia* strain obtain a fertility advantage because males infected with *Wolbachia* can cause female embry-onic lethality when they mate with uninfected females or females infected with a different *Wolbachia* strain [3,4]. Aside from these alterations that enhance the spread of *Wolbachia*, there can also be costs to the infection that slow its spread [5–7]. These effects of *Wolbachia* are normally based on investigations of natural *Wolbachia* strains that have existed in their hosts for many years, but are also frequently found in artificial infections [8].

In the last decade, several artificially introduced *Wolbachia* strains have been successfully used to reduce the transmission of arboviral diseases by *Aedes aegypti* mosquitoes [9,10], including field releases with the *Wolbachia* strains *w*Mel [11–13] and *w*AlbB [14]. Studies on native hosts tend to indicate only a weak fitness cost of *Wolbachia* [6], but *Wolbachia* can have diverse fitness costs in its newly introduced host *Ae. aegypti*, such as life-shortening, heat sensitivity, reduced quiescent egg viability and reduced blood feeding success [10,15–17]. The effects of *Wolbachia* in native and introduced hosts point to the possibility of a mutualistic relationship between *Wolbachia* developing over time because of co-evolution and adaptation. In fact, it is acknowledged that sometimes *Wolbachia* can act as both parasite and mutualist [5,18,19]. As a result, *Wolbachia* field release and establishment in new populations can be useful for studying evolutionary changes in *Wolbachia* given that the invasion history of the population is known [20–22].

In our previous study of the effects of *Wolbachia w*Mel and *w*AlbB on *Ae. aegypti* eggs, we discovered a novel reproductive effect induced by *w*AlbB, in which infected females that hatched from quiescent eggs were infertile despite successful mating and the proportion of females showing infertility was correlated with the duration of egg quiescence [23]. The *w*Mel strain also causes female infertility following egg storage, but to a lesser extent than *w*AlbB [23,24]. Several other reproductive fitness costs of *Wolbachia*, including those on the number and viability of eggs, can be attributed to the consequence of nutritional competition between *Wolbachia* and its host [25] and represent quantitative effects. However, in our study *Wolbachia*-infected females lost their fertility entirely, indicating a qualitative effect difficult to attribute to nutritional competition [23], so we suspect the involvement of other mechanisms.

In this paper, we investigate factors associated with *Ae. aegypti* female infertility induced by *Wolbachia* wAlbB infection after egg quiescence. Firstly, we tested if the effects of *Wolbachia* on female infertility are permanent and associated with a lack of fully developed ovaries. Secondly, we performed a larval starvation experiment to see if an extended larval period might also affect infertility. Thirdly, we undertook real-time PCR assays to understand the interaction between *Wolbachia* and the expression of genes influencing female development. Finally, we tested whether the loss of fully developed ovaries in infertile females affected the rate of female blood feeding. Our study highlights the novel nature of reproductive effects altered by

Wolbachia wAlbB in its new host *Ae. aegypti*, with implications for invasion of *Wolbachia* into uninfected populations and the evolution of mutualism more generally.

Methods

Ethics statement

The process of mosquito females feeding on human volunteers is approved by the University of Melbourne Human Ethics committee (approval 0723847). All adult subjects provided informed written consent (no children were involved).

Mosquito populations

We used uninfected and *Wolbachia* wAlbB-infected *Aedes aegypti*. The uninfected population was derived from eggs collected in 2019 from regions in Cairns, Queensland, Australia where *Ae. aegypti* were not infected with *Wolbachia*. The wAlbB-infected population was generated by microinjection [26] and infected females were crossed to males from the uninfected population regularly [27,28] to maintain a similar genetic background between populations. Mosquitoes were maintained in the laboratory following methods described previously [29] and all mosquito populations were screened routinely for their infection status. In the following experiments, non-stored wAlbB-infected and uninfected females were hatched from freshly-laid eggs that had been dried and conditioned for one week at 26 ± 1 °C, and stored wAlbB-infected in the same experiment had the same rearing and feeding conditions. In all experiments, female mosquitoes were blood fed on the same volunteer, and in each comparison, blood feeding of all cages was finished within two hours.

Determination of female fertility status

To determine the fertility status of individual females in the experiments, we hatched a mixture (approximately 1:1) of wAlbB-infected eggs that had been stored for two and four months, so that the proportion of infertile females is expected to be 25% and 80% respectively [23]. Female mosquitoes were blood fed on 3–5 days post-emergence. 200 engorged females were aspirated individually in 70 mL specimen cups with larval rearing water and sandpaper strips to allow them to lay eggs. After a week, we separated females that laid (fertile) or did not lay (infertile) eggs and grouped them separately in a *19.7-L BugDorm-1 adult cage (MegaView Science Co., Ltd., Taichung City, Xitun District, Taiwan*). This process is referred to as "fertility separation".

After "fertility separation", we measured the relative *Wolbachia* density of fertile and infertile females for groups of 16 individuals based on the $2^{\Delta Ct}$ method [30]. Screening based on real-time PCR assays followed methods described previously [31]. For each replicate group, samples were set up in a 384-well white plate and density measured by primers *aeg* and *wMwA* [32]; two consistent replicates (ΔCt <1) were obtained per group and their values were averaged for the density determination.

To test whether females scored as infertile based on the above criteria recovered fertility later in their lives or remained infertile, we then provided both fertile and infertile females with a second blood meal after the "fertility separation" process mentioned in the last paragraph, and isolated 20 "fertile" and 20 "infertile" females individually in 70 mL cups for another week to observe if their fertility status changed. We dissected the remaining females scored as "fertile" and "infertile" to examine the appearance of their ovaries. These females were killed by storing them at -80°C for 30 minutes then returned to room temperature for 10 minutes before dissecting them in saline solution (0.9% sodium chloride) under a compound light microscope (Motic B1 series, Australian Instrument Services Pty. Ltd., Australia).

To further examine fertility effects, we also dissected and examined ovaries in 65 wAlbBinfected females that had been stored as eggs for 11 weeks and that were 3–5 days post-emergence, but without a blood meal. These females had not been exposed to the "fertility separation" process but should show a high rate of infertility [23]. We used an NIS Elements BR imaging microscope (Nikon Instruments, Japan) to photograph the ovary tissues. The same dissections were also undertaken for a wAlbB transinfection line that had been repeatedly backcrossed to a Saudi Arabian background [33]. In this case we dissected ovaries from 113 wAlbB-infected mosquitoes with an egg stage that had lasted for 12 weeks, as well as 50 wAlbB-infected females with an egg stage that had lasted for 12 weeks (S1 Table).

Impact of larval starvation on mosquito infertility

To investigate whether the impact of Wolbachia infection on female infertility was only mediated through the egg stage or whether the prepupal stage had an impact as a whole, we deprived 2nd instar mosquito larvae of food for two weeks before feeding them again until pupation. We also set up non-starved controls where larvae were provided with food ad libi*tum.* Each starvation treatment was performed with both stored eggs (egg stage had lasted for 12 weeks) and non-stored eggs (egg stage had lasted for one week) for a total of four treatments to test if any fertility effects might accumulate across the larval and egg stages. In each treatment, we hatched 400 larvae and reared them in a tray containing 4 L of reverse osmosis water [29]. Before the blood meal, 15 individual females randomly selected from each treatment (3–5 days post-emergence) were screened for Wolbachia infection and Wolbachia density, using the screening methods described in the previous section. To test the proportion of females that were infertile, we set up replicates consisting of groups of 30 individual females emerging from the same rearing tray. For the wAlbB-infected line, we had two replicates for the treatment in which infected females were neither stored nor starved, and three replicates for the other three experimental treatments. Female infertility was tested through the "fertility separation" process. For the control uninfected Ae. aegypti line, we set up all four similar treatments and dissected 30 individuals at 3-5 days post-emergence from each treatment to check for the presence of ovaries (S1 Table).

Gene expression assays

We selected three genes related to mosquito reproduction and tested their expression levels in females at different developmental stages through a real-time PCR assay [31,32]. One of the genes tested was the *vitellogenin receptor* (*vgr*): following blood feeding, vitellogenin is secreted from the fat body and internalized by ovaries through the receptor VgR [34–36]. The other two genes were the *ecdysone receptor* (*ecr*) [37,38] and the *eggshell organizing factor* (*eof*); the vitellogenin transcript is regulated by *ecr* while *eof* is an essential gene encoding a protein for eggshell formation at the late stage of egg production [39] (S2 Table). Gene expression level was quantified relative to a control gene, *RPS17* [40]. We tested female pupae 0.5–1.5 days post-pupation and female adults 0.5–1.5 days post-emergence in *w*AlbB-infected females that hatched from stored eggs (egg stage had lasted for 14 weeks) and non-stored eggs (egg stage had lasted for one week) considered as the control. For the *w*AlbB-infected females (egg stage had lasted for one week) considered as the control. For the *w*AlbB-infected females after testing female fertility, and in this comparison fertile females were treated as the control. Mosquito samples were

given a second blood meal and stored in RNA*later* (Sigma Aldrich Cat No. R0901-100ML) on the third day after feeding, before RNA extraction, reverse transcription and real-time PCR assays were conducted. In all of the above comparisons, we screened 10 individuals from each group; the details of RNA extraction and real-time PCR can be found in S1 Text.

Blood feeding rate of female mosquitoes

Normally, fully engorged fertile females are reluctant to feed within a gonotrophic cycle. In this experiment, we tested egg storage impacts on subsequent feeding by females, sourcing mosquitoes from the same groups as used for the above gene expression assays. After determining female fertility, we measured the blood feeding rate of female mosquitoes by providing fertile and infertile females a third blood meal, three days after their second blood feeding, when engorged blood was almost digested. The same volunteer who provided previous blood meals fed the mosquitoes in a BugDorm cage for 15 minutes. Fully engorged females with blood visible in the abdomen were considered as having successfully fed after this feeding attempt. Three replicates were completed with 20–30 individual females for each replicate. We also weighed 20 random females that had fully fed and 20 that had not been provided with a blood meal from all three replicates to estimate their blood meal weight using a Sartorius Analytical balance BP 210 D (Sartorius, Gottigen, Germany, Readability: 0.01 mg).

Statistics

We used R v. 3.6.0 with R studio v. 1.1.453 to conduct data analyses and visualizations [41], using the "car" library for ANOVA [42], the base library for other statistical analyses, and the "ggplot2" library for visualization [43]. After real-time PCR measurement, we used the $2^{-\Delta\Delta Ct}$ method to compare gene expression levels [44], the values of $2^{-\Delta\Delta Ct}$ were natural log-transformed for ANOVA analysis, and a Tukey's honest significant difference (HSD) test [45] was used for further pairwise comparisons. For *Wolbachia* density we used the $2^{\Delta Ct}$ method [30] to calculate relatively density before log10 transformation for ANOVA analysis. For the proportion data from larval starvation and blood feeding experiments, we analysed data with binomial logistic regression models [46]. In the blood feeding rate test, student's t-tests were used to test for changes of mosquito weight after blood feeding. Treatments in this study are listed in S3 Table.

Results

Immature ovaries in infertile Aedes aegypti females

We performed a "fertility separation" for *w*AlbB-infected females and provided a second blood meal to fertile and infertile females. We confirmed that all 20 females we scored as fertile and the 20 we scored as infertile after the first gonotrophic cycle maintained their phenotype in the second gonotrophic cycle. We dissected and compared the ovarian appearance of fertile and infertile females and discovered that developed ovaries could not be observed in infertile females under a microscope (Fig 1). Only occasionally (around one in ten) can immature ovarian structure be seen, but with a narrow width similar to that of a Malpighian tube (S1 Fig). We also found that fertile females had higher relative densities of *Wolbachia* (S2 Fig: ANOVA comparing fertile and infertile females: $F_{1,30} = 8.632$, P = 0.006, fertile: mean \pm se = 5.15 \pm 0.906; infertile: mean \pm se = 2.290 \pm 0.673).

For the 3–5 days post-emergence mosquitoes dissected without separating them into fertile and infertile categories, we found that the majority of *w*AlbB females stored as eggs for 11–12 weeks lacked visible ovaries (S1 Table) in contrast to the uninfected population and females





that did not emerge from stored eggs. This pattern was also evident in the Saudi Arabian populations which had a different genetic background (S1 Table).

Impact of larval starvation on mosquito infertility

We deprived mosquito larvae of food to increase the duration of the larval stage from approximately one week to three weeks to test for any impact of larval starvation on infertility in *w*AlbB-infected females. We found significant effects of both storage and starvation treatment, with an interaction between these terms (Fig 2A: logistic regression testing effects of storage: $F_{1,7} = 256.304$, p < 0.001; starvation: $F_{1,7} = 32.412$, p < 0.001; interaction: $F_{1,7} = 21.515$, p = 0.002). Infertility associated with *w*AlbB infection increased further after larval starvation over and above that seen with stored eggs by 15%, and the extension of the larval stage by a period of starvation also induced infertility of around 10% by itself. For uninfected *Ae. aegypti* treated in the same manner, all females that were dissected contained ovaries, indicating the essential role of *Wolbachia w*AlbB in inducing female infertility during larval starvation and egg storage (Fig 2A and S1 Table).

Before blood feeding to identify female fertility status, we collected 15 females of 3–5 days post-emergence to screen for their *Wolbachia* infection status. We failed to detect an infection in five out of 60 individuals and these were excluded from the density analysis (S4 Table). *Wolbachia* density was significantly influenced by larval starvation but not egg quiescence, with a significant interaction (Fig 2B: ANOVA: starvation: $F_{1,51} = 7.145$, p = 0.010; egg quiescence: $F_{1,51} = 2.475$, p = 0.122, interaction between starvation and quiescence: $F_{1,51} = 8.396$, p = 0.006). It is likely that egg quiescence decreased *Wolbachia* density, but density increased with larval starvation, potentially cancelling out this effect [47].



Fig 2. (A) Boxplots showing the proportion of infertile females in *w*AlbB-infected and uninfected mosquitoes with an egg stage that had lasted for 12 weeks (stored) or one week and a larvae stage that had been deprived of food for two weeks (starved) or that had not been deprived of food. Mosquitoes that had not been treated with "stored" or "starved" are designated as "control". No uninfected females from all four treatments were infertile. Values above boxplots represent corresponding averaged percentages based on two replicates for controls and three replicates for treatments with each replicate containing 30 individuals. (B) Boxplots of relative *Wolbachia* density in 3–5 days post-emergence females with unknown fertility status. Each group is based on two consistent real-time PCR replicates.

Gene expression assays

We selected three genes related to reproductive development and tested their expression level at different developmental stages. At the pupal stage, there were significant differences on gene expression among the three mosquito lines (wAlbB-infected with an egg stage that had lasted for 14 weeks; wAlbB-infected and *Wolbachia* uninfected with an egg stage that had lasted for only one week (non-stored)), but no interaction between lines and the genes or overall difference between the genes (Fig 3A, ANOVA on relative expression: mosquito group: $F_{2,81} = 26.717$, p < 0.001; genes: $F_{2,81} = 1.044$, p = 0.357; interaction: $F_{4,81} = 0.423$, p = 0.792). Specifically, in Tukey's HSD tests, wAlbB-infected females hatched from stored eggs had lower expression levels for all three genes than uninfected females, while only *ecr* had a lower expression level in *w*AlbB-infected females that had been stored compared to infected females that not been stored (Fig 3A).

For young adults 0.5–1.5 days post-emergence, significant differences in gene expression levels were found among mosquito groups, but there was no interaction with the genes (Fig 3B, ANOVA on relative gene expression: mosquito group: $F_{2,81} = 15.641$, p < 0.001; genes: $F_{2,81} = 0.383$, p = 0.683; interaction: $F_{4,81} = 7.100$, p = 0.792). Specifically, only differences for *vgr* expression were significant between *w*AlbB-infected females that been stored and the other two mosquito groups, with some individuals from stored eggs having very low expression levels—these females are probably infertile (Fig 3B).

After "fertility separation", female mosquitoes were provided a second blood meal and were screened for gene expression levels three days later. Significant differences were found between fertile and infertile females, and also between genes and their interaction (Fig 4, ANOVA on relative gene expression: mosquito group: $F_{1,54} = 235.696$, p < 0.001; genes: $F_{2,54} = 68.553$,



Fig 3. Boxplots of relative expression of reproduction-related genes *ecdysone receptor (ecr)*, *eggshell organizing factor (eof)* and *vitellogenin receptor (vgr)* in female (A) pupae and (B) young adults (1 ± 0.5 days post-emergence). These three genes were normalized to reference gene *RPS17*. Values with the same letter are not significantly different based on Tukey's HSD tests (<u>S5 Table</u>). Results are based on two consistent real-time PCR replicates of ten individual females from each group.

p < 0.001; interaction: $F_{2,54} = 68.553$, p < 0.001). Genes *eof* and *vgr* showed significant differences between fertile and infertile females for expression, especially in the case of *vgr* whose expression in infertile mosquitoes was only around 0.01% of that seen in fertile mosquitoes (Fig 4), likely reflecting the fact that infertile females had incomplete ovarian formation.



Fig 4. Boxplots of relative expression of reproduction-related genes *ecdysone receptor (ecr), eggshell organizing factor (eof)* and *vitellogenin receptor (vgr)* in fertile and infertile females one week after their second blood meal. Genes were normalized to reference gene *RPS17.* Values with the same letter are not significantly different according to Tukey's HSD tests. Results are based on two consistent real-time PCR replicates of ten individual females from each group.

https://doi.org/10.1371/journal.pntd.0010913.g004



Fig 5. Boxplots of (A) female proportion that blood fed three days after they had been fully fed, in which three replicates were completed with 20–30 individual females for each replicate; and (B) the weights of fed and unfed females. 20 females that had fully fed and 20 that had not been provided with a blood meal from all three replicates were weighed at random.

Blood feeding rate

After "fertility separation", we provided female mosquitoes a third blood meal on the third day after they had become engorged with their second blood meal to test their blood feeding behaviour. Infertile females had a higher proportion feeding compared to fertile and uninfected females (Fig 5A, logistic regression, $F_{2,6} = 61.037$, p < 0.001). We also compared the weight of fully fed and unfed females and noted significant differences between mosquito groups, feeding status and their interaction (Fig 5B, ANOVA: mosquito group: $F_{1,73} = 29.06$, p < 0.001; feeding: $F_{1,73} = 322.89$, p < 0.001; interaction: $F_{2,73} = 15.26$, p < 0.001). Specifically, significant weight increases were recorded for both *w*AlbB-infected fertile females ($t_{35} = 9.627$, p < 0.001) and *w*AlbB-infected infertile females ($t_{38} = 15.658$, p < 0.001), but the weight of fully fed fertile females was significantly lower compared to infertile females ($t_{35} = -4.645$, p < 0.001), while there was no significant difference between unfed fertile and infertile females ($t_{38} = -1.096$, p = 0.280), suggesting that infertile females took in a larger amount of blood during successive feeding periods.

Discussion

Wolbachia is a well-known bacterium that can alter the reproduction of its host to benefit its spread. In our previous study, *Wolbachia* wAlbB was found to cause *Ae. aegypti* females to become infertile following egg storage [23], inhibiting vertical transmission. In this paper, we investigated this phenomenon and obtained the following results: 1) in infertile females, the development of ovaries has been interrupted by *Wolbachia*, which is supported by ovarian dissection and gene expression assays; 2) the effect of *Wolbachia* on female fertility accumulates across pre-pupal stages, in that the frequency of female infertility depends on the duration before metamorphosis (including egg and larval stages); and 3) infertile females maintain other female characteristics such as mating and blood feeding, but they do not enter a gonotrophic cycle and blood feeding occurs more frequently. We, therefore, confirmed a novel reproductive alteration of *Wolbachia* wAlbB in its new host, *Ae. aegypti* when coupled with an extended pre-pupal stage.

The effect of Wolbachia on female infertility is different from other fitness impacts of Wolbachia and the four typical reproductive alterations (parthenogenesis, feminization, male killing and cytoplasmic incompatibility) in which Wolbachia alters host reproduction in ways that increase its vertical transmission success [4,48]. The infertility effect we describe here converts functional females into non-functional females without male characteristics, suggesting different Wolbachia-related mechanisms to those involved in other phenomena like feminization [49,50]. By testing the expression level of vgr, ecr and eof at the female pupal, and adult stages before and after blood feeding, we found substantial differences in gene expression between fertile and infertile groups. For pupae, large variation in expression levels were found among individuals, especially for expression of the gene vgr, probably reflecting the fact that its expression mainly occurs in a narrow period during the pupal stage. wAlbB-infected Ae. aegypti females that had been stored long-term as eggs had lower expression levels of the three reproductive genes compared with uninfected females, especially for the gene vgr at the young adult stage, suggesting the formation and development of oocytes was impacted. This is also supported by the very low relative expression of vgr and eof in infertile females when compared with fertile females at the third day after blood feeding, and the absence of mature ovaries in infertile females. The lower overall Wolbachia densities in infertile females is also consistent with previous work showing high wAlbB densities in ovaries (S2 Fig) [51]. Moreover, we found much lower expression of ecr in wAlbB-infected female pupae hatched from long-stored eggs, while this difference was not found in young adults. As ecr is expressed in a variety of organs to encode the receptor of a hormone [52], the down-regulation of ecr may indicate that egg quiescence has a substantial impact on the pupation process of wAlbB-infected females. However, interactions between Wolbachia and specific reproductive signals and mechanisms are poorly understood and require further investigation for example, through hormone level tests, transcriptome analysis and pathway analysis [53,54].

In population replacement approaches [8], the novel reproductive alteration induced by *Wolbachia wAlbB* could slow the spread of the infection in mosquito populations as females are less likely to produce offspring. This may occur not only when the egg desiccation period is extended during dry and warm seasons [55,56], but also when the larval period is extended due to poor provision of food [57–59]. These effects can reduce the efficiency of *Wolbachia* invasion which aims to inhibit the transmission of arboviruses [8,28,60]. While *w*AlbB has established successfully in Malaysia where the warm and humid year-round climate precludes extended egg desiccation periods and promotes fast development of larvae [14,61], our discovery provides guidance for future releases in other climates and highlights the importance of future monitoring [23]. Our results also have implications for laboratory studies and release programs where it is common to store large numbers of eggs prior to experiments and transportation to release sites.

Our observation that infertile females show an increased rate of blood feeding highlights a potential risk of increased nuisance biting following a *Wolbachia* release program, which may lead to community discontentment. Normal *Ae. aegypti* females ingest human blood to obtain unique nutrients for egg development [62,63]. However, while infertile females lack ovaries, they still blood feed. Previous research found that multiple feeding within a gonotrophic cycle is caused by nutritional reserve depletion or feeding interruption [64]. Nevertheless, we found a large proportion of infertile females fed again on the third day after their second blood meal, indicating they do not follow the same gonotrophic cycle. It is unclear if and how the process of blood digestion is impacted by infertility. Infertile females also took in a larger amount of blood, though this might be a compensatory reaction towards starvation at egg and larval stages. This behavioural change of *Wolbachia*-infected infertile females has significant implications for vector control population release strategies [8,65]; in particular it would be interesting

to test whether the ability of these infertile females to carry and transmit arbovirus has changed as the density of their *Wolbachia* infection may be lower than that of fertile females [66,67]. Similar concerns have been raised about increased biting frequencies of irradiated female mosquitoes in sterile insect technique (SIT) programs [68], however in population replacement approaches the prevalence of infertile females is likely to be much higher in *Wolbachia*-infected mosquito populations compared to in SIT irradiated females that are only released accidentally.

The reproductive alterations of Wolbachia discovered in this study have evolutionary implications. So far, our work relates to Ae. aegypti, where the Wolbachia wAlbB was artificially introduced recently from a close relative of Ae. aegypti, Aedes albopictus, in order to reduce the transmission of arboviral diseases [69]. A Wolbachia strain may induce similar fitness costs in a new host when compared to its native host, such as in the case of life-shortening caused by wMelPop which is expressed in both Drosophila melanogaster and Ae. aegypti [10], and remains stably expressed following long-term laboratory culture [70]. We do not yet know if there are similar reproductive effects of Wolbachia wAlbB expressed in its native host, Ae. albopictus. On the other hand, we have also discovered a smaller loss of female infertility in wMel infected Ae. aegypti [23], though the native host of wMel, D. melanogaster, does not enter egg quiescence, although there is a reduced fecundity of *w*Mel-infected females under dormancy conditions [71]. Although other Wolbachia strains remain to be investigated, it is possible that Ae. aegypti has mechanisms underlying ovarian formation that can be interrupted by endosymbionts. However, it is unclear why female infertility is observed in Wolbachia-infected mosquitoes after increased periods of egg storage or larval development time, and to what extent the duration of pre-pupal stage that causes female infertile will be impacted by mosquito genetic backgrounds. There may be cumulative effects of Wolbachia at early stages of development contributing to failure of ovarian formation. One possible hypothesis which requires further investigation is that Wolbachia competes for some essential but sparse nutrients with its host Ae. aegypti at an early life stage, such as amino acids and cholesterol that mainly come from human blood [72,73].

In conclusion, we further investigated a novel reproductive alteration of *Wolbachia* that we discovered previously [23]. We confirmed that *Ae. aegypti* females infected with *Wolbachia w*AlbB can become infertile when they are unable to form functional ovaries during metamorphosis, but these females retain other female characteristics, leading to an increased biting frequency. Our study provides significant guidance for future *Wolbachia* releases and has important evolutionary implications for understanding the reproductive alteration of *Wolbachia*, especially in novel hosts.

Supporting information

S1 Table. Summary of the number of mosquitoes dissected without "fertility separation" and their ovarian developmental status. (DOCX)

S2 Table. Primers used to measure the expression of three essential reproductive-related genes in *Aedes aegypti* females. (DOCX)

S3 Table. List of treatments and their definitions. (DOCX)

S4 Table. Summary of the number of uninfected mosquitoes that were excluded in the density analysis in the *Aedes aegypti* larval starvation experiment. (DOCX) S5 Table. Density means and results of posthoc tests distinguishing the groups (different letters indicate significant differences between the means) based on data presented in Figs 3 and 4.

(DOCX)

S1 Text. Methods for expression analysis. (DOCX)

S1 Fig. An uncommon case where immature ovarian structures can be seen in infertile females, with the width of ovaries similar to Malpighian tubules. (DOCX)

S2 Fig. Box plots of relative *Wolbachia* density of fertile and infertile *w*AlbB-infected females after females were separated into fertile and infertile groups (one week after blood meal).

(DOCX)

S3 Fig. Original pictures for Fig 1. (DOCX)

Acknowledgments

We thank Véronique Paris for assistance in weighing mosquitoes.

Author Contributions

Conceptualization: Meng-Jia Lau, Perran A. Ross, Ary A. Hoffmann.

Data curation: Meng-Jia Lau, Perran A. Ross, Ary A. Hoffmann.

Formal analysis: Meng-Jia Lau, Perran A. Ross.

Funding acquisition: Ary A. Hoffmann.

Investigation: Meng-Jia Lau, Perran A. Ross, Nancy M. Endersby-Harshman.

Methodology: Meng-Jia Lau, Perran A. Ross, Nancy M. Endersby-Harshman, Qiong Yang, Ary A. Hoffmann.

Project administration: Ary A. Hoffmann.

Resources: Ary A. Hoffmann.

Supervision: Perran A. Ross, Nancy M. Endersby-Harshman, Qiong Yang, Ary A. Hoffmann.

Validation: Nancy M. Endersby-Harshman, Qiong Yang.

Visualization: Meng-Jia Lau.

Writing - original draft: Meng-Jia Lau, Ary A. Hoffmann.

Writing – review & editing: Meng-Jia Lau, Perran A. Ross, Nancy M. Endersby-Harshman, Qiong Yang, Ary A. Hoffmann.

References

 Zug R, Hammerstein P. Still a host of hosts for Wolbachia: analysis of recent data suggests that 40% of terrestrial arthropod species are infected. PLoS One. 2012; 7(6):e38544. Epub 2012/06/12. https://doi. org/10.1371/journal.pone.0038544 PMID: 22685581; PubMed Central PMCID: PMC3369835.

- Stouthamer R, Breeuwer JA, Hurst GD. Wolbachia pipientis: microbial manipulator of arthropod reproduction. Annu Rev Microbiol. 1999; 53:71–102. Epub 1999/11/05. <u>https://doi.org/10.1146/annurev.</u> micro.53.1.71 PMID: 10547686.
- Werren JH, Baldo L, Clark ME. Wolbachia: master manipulators of invertebrate biology. Nat Rev Microbiol. 2008; 6(10):741.
- 4. O'Neill SL, Hoffmann A, Werren J. Influential passengers: inherited microorganisms and arthropod reproduction: Oxford University Press; 1997.
- Zug R, Hammerstein P. Bad guys turned nice? A critical assessment of *Wolbachia* mutualisms in arthropod hosts. Biol Rev Camb Philos Soc. 2015; 90(1):89–111. Epub 2014/03/13. https://doi.org/10.1111/ brv.12098 PMID: 24618033.
- 6. Hoffmann AA, Turelli M, Harshman LG. Factors affecting the distribution of cytoplasmic incompatibility in *Drosophila simulans*. Genetics. 1990; 126(4):933–48. WOS:A1990EL33400014.
- Perrot-Minnot MJ, Cheval B, Migeon A, Navajas M. Contrasting effects of *Wolbachia* on cytoplasmic incompatibility and fecundity in the haplodiploid mite *Tetranychus urticae*. J Evol Biol. 2002; 15(5):808– 17.
- Ross PA, Turelli M, Hoffmann AA. Evolutionary ecology of *Wolbachia* releases for disease control. Annu Rev Genet. 2019; 53:93–116. Epub 2019/09/11. <u>https://doi.org/10.1146/annurev-genet-112618-043609</u> PMID: 31505135.
- Ogunlade ST, Adekunle AI, Meehan MT, Rojas DP, McBryde ES. Modeling the potential of wAu-Wolbachia strain invasion in mosquitoes to control Aedes-borne arboviral infections. Sci Rep. 2020; 10 (1):16812. Epub 20201008. <u>https://doi.org/10.1038/s41598-020-73819-1</u> PMID: <u>33033285</u>; PubMed Central PMCID: PMC7544821.
- McMeniman CJ, Lane RV, Cass BN, Fong AWC, Sidhu M, Wang Y-F, et al. Stable introduction of a lifeshortening *Wolbachia* infection into the mosquito *Aedes aegypti*. Science. 2009; 323(5910):141–4. https://doi.org/10.1126/science.1165326 WOS:000262104100061. PMID: 19119237
- Ryan PA, Turley AP, Wilson G, Hurst TP, Retzki K, Brown-Kenyon J, et al. Establishment of wMel Wolbachia in Aedes aegypti mosquitoes and reduction of local dengue transmission in Cairns and surrounding locations in northern Queensland, Australia. Gates Open Res. 2019; 3:1547. https://doi.org/10. 12688/gatesopenres.13061.2 PMID: 31667465
- Tantowijoyo W, Andari B, Arguni E, Budiwati N, Nurhayati I, Fitriana I, et al. Stable establishment of wMel Wolbachia in Aedes aegypti populations in Yogyakarta, Indonesia. PLoS Negl Trop Dis. 2020; 14 (4):e0008157. Epub 2020/04/18. https://doi.org/10.1371/journal.pntd.0008157 PMID: 32302295; PubMed Central PMCID: PMC7190183.
- Gesto JSM, Ribeiro GS, Rocha MN, Dias FBS, Peixoto J, Carvalho FD, et al. Reduced competence to arboviruses following the sustainable invasion of Wolbachia into native Aedes aegypti from Southeastern Brazil. Sci Rep. 2021; 11(1):10039. Epub 20210511. https://doi.org/10.1038/s41598-021-89409-8 PMID: 33976301; PubMed Central PMCID: PMC8113270.
- Nazni WA, Hoffmann AA, NoorAfizah A, Cheong YL, Mancini MV, Golding N, et al. Establishment of Wolbachia strain wAlbB in Malaysian populations of Aedes aegypti for dengue control. Curr Biol. 2019; 29(24):4241–8.
- Turley AP, Moreira LA, O'Neill SL, McGraw EA. Wolbachia infection reduces blood-feeding success in the dengue fever mosquito, Aedes aegypti. PLoS Negl Trop Dis. 2009; 3(9):e516. <u>https://doi.org/10. 1371/journal.pntd.0000516</u> PMID: 19753103
- Yeap HL, Mee P, Walker T, Weeks AR, O'Neill SL, Johnson P, et al. Dynamics of the "Popcorn" Wolbachia infection in outbred Aedes aegypti informs prospects for mosquito vector control. Genetics. 2011; 187(2):583–95. https://doi.org/10.1534/genetics.110.122390 PMID: 21135075
- Ware-Gilmore F, Sgrò CM, Xi Z, Dutra HLC, Jones MJ, Shea K, et al. Microbes increase thermal sensitivity in the mosquito Aedes aegypti, with the potential to change disease distributions. PLoS Negl Trop Dis. 2021; 15(7):e0009548. Epub 2021/07/23. https://doi.org/10.1371/journal.pntd.0009548 PMID: 34292940; PubMed Central PMCID: PMC8297775.
- Fenn K, Blaxter M. Wolbachia genomes: revealing the biology of parasitism and mutualism. Trends Parasitol. 2006; 22(2):60–5. Epub 2006/01/13. <u>https://doi.org/10.1016/j.pt.2005.12.012</u> PMID: 16406333.
- 19. Weeks AR, Turelli M, Harcombe WR, Reynolds KT, Hoffmann AA. From parasite to mutualist: rapid evolution of *Wolbachia* in natural populations of *Drosophila*. PLoS Biol. 2007; 5(5):e114.
- Lau MJ, Schmidt TL, Yang Q, Chung J, Sankey L, Ross PA, et al. Genetic stability of *Aedes aegypti* populations following invasion by *w*Mel *Wolbachia*. BMC Genomics. 2021; 22(1):894. Epub 2021/12/ 16. https://doi.org/10.1186/s12864-021-08200-1 PMID: <u>34906084</u>; PubMed Central PMCID: PMC8670162.

- Carrington LB, Hoffmann AA, Weeks AR. Monitoring long-term evolutionary changes following Wolbachia introduction into a novel host: the Wolbachia popcorn infection in Drosophila simulans. Proc Biol Sci. 2010; 277(1690):2059–68.
- Ford SA, Allen SL, Ohm JR, Sigle LT, Sebastian A, Albert I, et al. Selection on Aedes aegypti alters Wolbachia-mediated dengue virus blocking and fitness. Nat Microbiol. 2019; 4(11):1832–9. Epub 2019/08/ 26. https://doi.org/10.1038/s41564-019-0533-3 PMID: 31451771.
- Lau MJ, Ross PA, Hoffmann AA. Infertility and fecundity loss of Wolbachia-infected Aedes aegypti hatched from quiescent eggs is expected to alter invasion dynamics. PLoS Negl Trop Dis. 2021; 15(2): e0009179. Epub 2021/02/17. https://doi.org/10.1371/journal.pntd.0009179 PMID: 33591971; PubMed Central PMCID: PMC7909672.
- 24. Ross PA, Hoffmann AA. Local adaptation of Aedes aegypti mosquitoes to Wolbachia-induced fitness costs. bioRxiv. 2022.
- Allman MJ, Fraser JE, Ritchie SA, Joubert DA, Simmons CP, Flores HA. Wolbachia's deleterious impact on Aedes aegypti egg development: The potential role of nutritional parasitism. Insects. 2020; 11(11):735. Epub 2020/10/31. https://doi.org/10.3390/insects11110735 PMID: 33120915.
- Xi ZY, Khoo CCH, Dobson SL. Wolbachia establishment and invasion in an Aedes aegypti laboratory population. Science. 2005; 310(5746):326–8. https://doi.org/10.1126/science.1117607 WOS:000232670100056. PMID: 16224027
- Garcia GdA, Sylvestre G, Aguiar R, da Costa GB, Martins AJ, Lima JBP, et al. Matching the genetics of released and local *Aedes aegypti* populations is critical to assure *Wolbachia* invasion. PLoS Negl Trop Dis. 2019; 13(1):e0007023. https://doi.org/10.1371/journal.pntd.0007023 PMID: 30620733
- Hoffmann AA, Montgomery B, Popovici J, Iturbe-Ormaetxe I, Johnson P, Muzzi F, et al. Successful establishment of *Wolbachia* in *Aedes* populations to suppress dengue transmission. Nature. 2011; 476 (7361):454–7.
- Ross PA, Axford JK, Richardson KM, Endersby-Harshman NM, Hoffmann AA. Maintaining Aedes aegypti mosquitoes infected with Wolbachia. J Vis Exp. 2017; 14(126):e56124. <u>https://doi.org/10.3791/</u> 56124 PMID: 28829414
- Unckless RL, Boelio LM, Herren JK, Jaenike J. Wolbachia as populations within individual insects: causes and consequences of density variation in natural populations. Proc Biol Sci. 2009; 276 (1668):2805–11. Epub 2009/05/08. https://doi.org/10.1098/rspb.2009.0287 PMID: 19419989; PubMed Central PMCID: PMC2839946.
- Lee SF, White VL, Weeks AR, Hoffmann AA, Endersby NM. High-throughput PCR assays to monitor Wolbachia infection in the dengue mosquito (Aedes aegypti) and Drosophila simulans. Appl Environ Microbiol. 2012; 78(13):4740–3. https://doi.org/10.1128/AEM.00069-12 PMID: 22522691
- Lau M-J, Hoffmann AA, Endersby-Harshman NM. A diagnostic primer pair to distinguish between wMel and wAlbB Wolbachia infections. PLoS One. 2021; 16(9):e0257781. <u>https://doi.org/10.1371/journal.pone.0257781</u> PMID: 34555085
- Ross PA, Gu X, Robinson KL, Yang Q, Cottingham E, Zhang Y, et al. A wAlbB Wolbachia Transinfection Displays Stable Phenotypic Effects across Divergent Aedes aegypti Mosquito Backgrounds. Appl Environ Microbiol. 2021; 87(20):e0126421. Epub 2021/08/12. https://doi.org/10.1128/aem.01264-21 PMID: 34379518; PubMed Central PMCID: PMC8478461.
- Raikhel AS, Lea AO. Previtellogenic development and vitellogenin synthesis in the fat body of a mosquito: an ultrastructural and immunocytochemical study. Tissue Cell. 1983; 15(2):281–99. Epub 1983/ 01/01. https://doi.org/10.1016/0040-8166(83)90023-x PMID: 6349013.
- Sappington TW, Raikhel AS. Molecular characteristics of insect vitellogenins and vitellogenin receptors. Insect Biochem Mol Biol. 1998; 28(5–6):277–300. Epub 1998/08/06. https://doi.org/10.1016/s0965-1748(97)00110-0 PMID: 9692232.
- Tufail M, Takeda M. Insect vitellogenin/lipophorin receptors: molecular structures, role in oogenesis, and regulatory mechanisms. J Insect Physiol. 2009; 55(2):87–103. Epub 2008/12/17. https://doi.org/10. 1016/j.jinsphys.2008.11.007 PMID: 19071131.
- Hodin J, Riddiford LM. The ecdysone receptor and ultraspiracle regulate the timing and progression of ovarian morphogenesis during *Drosophila metamorphosis*. Dev Genes Evol. 1998; 208(6):304–17. Epub 1998/08/26. https://doi.org/10.1007/s004270050186 PMID: 9716721.
- Martín D, Wang SF, Raikhel AS. The vitellogenin gene of the mosquito Aedes aegypti is a direct target of ecdysteroid receptor. Mol Cell Endocrinol. 2001; 173(1–2):75–86. Epub 2001/02/27. <u>https://doi.org/ 10.1016/s0303-7207(00)00413-5 PMID: 11223179</u>.
- Isoe J, Koch LE, Isoe YE, Rascón AA Jr., Brown HE, Massani BB, et al. Identification and characterization of a mosquito-specific eggshell organizing factor in *Aedes aegypti* mosquitoes. PLoS Biol. 2019; 17 (1):e3000068. Epub 2019/01/09. https://doi.org/10.1371/journal.pbio.3000068 PMID: 30620728; PubMed Central PMCID: PMC6324781.

- 40. Dzaki N, Ramli KN, Azlan A, Ishak IH, Azzam G. Evaluation of reference genes at different developmental stages for quantitative real-time PCR in Aedes aegypti. Sci Rep. 2017; 7:43618. Epub 2017/03/17. https://doi.org/10.1038/srep43618 PMID: 28300076; PubMed Central PMCID: PMC5353741.
- 41. Team RC. R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria. 2019.
- 42. Fox J, Weisberg S, Adler D, Bates D, Baud-Bovy G, Ellison S, et al. Package 'car'. Vienna: R Foundation for Statistical Computing. 2012.
- **43.** Wickham H. ggplot2: elegant graphics for data analysis: Springer; 2016.
- **44.** Livak KJ, Schmittgen TD. Analysis of relative gene expression data using real-time quantitative PCR and the 2(-delta delta C(T)) method. Methods. 2001; 25(4):402–8. Epub 2002/02/16. <u>https://doi.org/10.1006/meth.2001.1262</u> PMID: <u>11846609</u>.
- **45.** Tukey JW. Comparing individual means in the analysis of variance. Biometrics. 1949; 5(2):99–114. Epub 1949/06/01. PMID: <u>18151955</u>.
- Hastie TJ. Generalized additive models. In: Science S, editor. Statistical models in S. 1: Routledge; 2017. p. 249–307.
- Lau MJ, Ross PA, Endersby-Harshman NM, Hoffmann AA. Impacts of low temperatures on Wolbachia (Rickettsiales: Rickettsiaceae)-infected Aedes aegypti (Diptera: Culicidae). J Med Entomol. 2020; 57 (5):1567–74. Epub 2020/04/21. https://doi.org/10.1093/jme/tjaa074 PMID: 32307514.
- 48. Werren JH. Biology of Wolbachia. Annu Rev Entomol. 1997; 42:587-609.
- **49.** Rigaud T, Juchault P, Mocquard JP. The evolution of sex determination in isopod crustaceans. Bioessays. 1997; 19(5):409–16.
- Bandi C, Dunn AM, Hurst GD, Rigaud T. Inherited microorganisms, sex-specific virulence and reproductive parasitism. Trends Parasitol. 2001; 17(2):88–94. Epub 2001/03/03. https://doi.org/10.1016/ s1471-4922(00)01812-2 PMID: 11228015.
- Joubert DA, Walker T, Carrington LB, De Bruyne JT, Kien DHT, Hoang NLT, et al. Establishment of a Wolbachia superinfection in Aedes aegypti mosquitoes as a potential approach for future resistance management. PLoS Pathog. 2016; 12(2):e1005434. <u>https://doi.org/10.1371/journal.ppat.1005434</u> PMID: 26891349
- Koelle MR, Talbot WS, Segraves WA, Bender MT, Cherbas P, Hogness DS. The *Drosophila EcR* gene encodes an ecdysone receptor, a new member of the steroid receptor superfamily. Cell. 1991; 67 (1):59–77. Epub 1991/10/04. https://doi.org/10.1016/0092-8674(91)90572-g PMID: 1913820.
- Belles X, Piulachs MD. Ecdysone signalling and ovarian development in insects: from stem cells to ovarian follicle formation. Biochim Biophys Acta. 2015; 1849(2):181–6. Epub 2014/06/19. https://doi. org/10.1016/j.bbagrm.2014.05.025 PMID: 24939835.
- Dansereau DA, Lasko P. The development of germline stem cells in Drosophila. Methods Mol Biol. 2008; 450:3–26. Epub 2008/03/29. https://doi.org/10.1007/978-1-60327-214-8_1 PMID: <u>18370048</u>; PubMed Central PMCID: PMC2729445.
- 55. Juliano SA O'Meara GF, Morrill JR, Cutwa MM. Desiccation and thermal tolerance of eggs and the coexistence of competing mosquitoes. Oecologia. 2002; 130(3):458–69. https://doi.org/10.1007/s004420100811 PMID: 20871747
- 56. Faull KJ, Williams CR. Intraspecific variation in desiccation survival time of Aedes aegypti (L.) mosquito eggs of Australian origin. J Vector Ecol. 2015; 40(2):292–300. Epub 2015/11/28. <u>https://doi.org/10.1111/jvec.12167</u> PMID: 26611964.
- Ross PA, Endersby NM, Hoffmann AA. Costs of three Wolbachia infections on the survival of Aedes aegypti larvae under starvation conditions. PLoS Negl Trop Dis. 2016; 10(1):e0004320. https://doi.org/ 10.1371/journal.pntd.0004320 PMID: 26745630
- Barrera R, Medialdea V. Development time and resistance to starvation of mosquito larvae. J Nat Hist. 1996; 30(3):447–58.
- Barrera R Competition and resistance to starvation in larvae of container-inhabiting Aedes mosquitoes. Ecological Entomology. 1996; 21(2):117–27.
- Koh C, Audsley MD, Di Giallonardo F, Kerton EJ, Young PR, Holmes EC, et al. Sustained Wolbachiamediated blocking of dengue virus isolates following serial passage in Aedes aegypti cell culture. Virus Evol. 2019; 5(1):vez012.
- Ahmad NA, Mancini MV, Ant TH, Martinez J, Kamarul GMR, Nazni WA, et al. Wolbachia strain wAlbB maintains high density and dengue inhibition following introduction into a field population of Aedes aegypti. Philos Trans R Soc Lond B Biol Sci. 2021; 376(1818):20190809. Epub 2020/12/29. https://doi. org/10.1098/rstb.2019.0809 PMID: 33357050; PubMed Central PMCID: PMC7776933.

- Gulia-Nuss M, Elliot A, Brown MR, Strand MR. Multiple factors contribute to anautogenous reproduction by the mosquito Aedes aegypti. J Insect Physiol. 2015; 82:8–16. Epub 2015/08/11. https://doi.org/10. 1016/j.jinsphys.2015.08.001 PMID: 26255841; PubMed Central PMCID: PMC4630150.
- Scott TW, Naksathit A, Day JF, Kittayapong P, Edman JD. A fitness advantage for *Aedes aegypti* and the viruses it transmits when females feed only on human blood. Am J Trop Med Hyg. 1997; 57(2):235– 9. Epub 1997/08/01. https://doi.org/10.4269/ajtmh.1997.57.235 PMID: 9288822.
- Farjana T, Tuno N. Multiple blood feeding and host-seeking behavior in Aedes aegypti and Aedes albopictus (Diptera: Culicidae). J Med Entomol. 2013; 50(4):838–46. Epub 2013/08/10. <u>https://doi.org/10.</u> 1603/me12146 PMID: 23926783.
- Almeida L, Duprez MG, Privat Y, Vauchelet N. Mosquito population control strategies for fighting against arboviruses. Math Biosci Eng. 2019; 16(6):6274–97. Epub 2019/11/09. <u>https://doi.org/10.3934/ mbe.2019313</u> PMID: 31698562.
- Johnson KN. The impact of Wolbachia on virus infection in mosquitoes. Viruses. 2015; 7(11):5705–17. Epub 2015/11/12. <u>https://doi.org/10.3390/v7112903</u> PMID: <u>26556361</u>; PubMed Central PMCID: PMC4664976.
- Terradas G, McGraw EA. Wolbachia-mediated virus blocking in the mosquito vector Aedes aegypti. Curr Opin Insect Sci. 2017; 22:37–44. Epub 2017/08/15. https://doi.org/10.1016/j.cois.2017.05.005 PMID: 28805637.
- Moretti R, Lampazzi E, Damiani C, Fabbri G, Lombardi G, Pioli C, et al. Increased biting rate and decreased Wolbachia density in irradiated Aedes mosquitoes. Parasit Vectors. 2022; 15(1):67. Epub 2022/02/26. https://doi.org/10.1186/s13071-022-05188-9 PMID: 35209944; PubMed Central PMCID: PMC8867665.
- 69. Xi Z, Dean JL, Khoo C, Dobson SL. Generation of a novel Wolbachia infection in Aedes albopictus (Asian tiger mosquito) via embryonic microinjection. Insect Biochem Mol Biol. 2005; 35(8):903–10. https://doi.org/10.1016/j.ibmb.2005.03.015 PMID: 15944085
- Ross PA, Axford JK, Callahan AG, Richardson KM, Hoffmann AA. Persistent deleterious effects of a deleterious Wolbachia infection. PLoS Negl Trop Dis. 2020; 14(4):e0008204–e. <u>https://doi.org/10.1371/journal.pntd.0008204</u> PMID: 32243448.
- Kriesner P, Conner WR, Weeks AR, Turelli M, Hoffmann AA. Persistence of a *Wolbachia* infection frequency cline in *Drosophila melanogaster* and the possible role of reproductive dormancy. Evolution. 2016; 70(5):979–97. https://doi.org/10.1111/evo.12923 WOS:000378946600003. PMID: 27076356
- 72. Geoghegan V, Stainton K, Rainey SM, Ant TH, Dowle AA, Larson T, et al. Perturbed cholesterol and vesicular trafficking associated with dengue blocking in Wolbachia-infected Aedes aegypti cells. Nat Commun. 2017; 8(1):526. Epub 2017/09/15. https://doi.org/10.1038/s41467-017-00610-8 PMID: 28904344; PubMed Central PMCID: PMC5597582.
- 73. Hansen IA, Attardo GM, Park JH, Peng Q, Raikhel AS. Target of rapamycin-mediated amino acid signaling in mosquito anautogeny. Proc Natl Acad Sci U S A. 2004; 101(29):10626–31. Epub 2004/07/02. https://doi.org/10.1073/pnas.0403460101 PMID: 15229322; PubMed Central PMCID: PMC489984.